



University of Groningen

Price leadership and unequal market sharing

Dijkstra, Pieter

IMPORTANT NOTE: You are advised to consult the publisher's version (publisher's PDF) if you wish to cite from it. Please check the document version below.

Document Version Publisher's PDF, also known as Version of record

Publication date: 2015

Link to publication in University of Groningen/UMCG research database

Citation for published version (APA): Dijkstra, P. (2015). Price leadership and unequal market sharing: collusion in experimental markets. (SOM Résearch Reports; Vol. 15005-EEF). University of Groningen, SOM research school.

Copyright

Other than for strictly personal use, it is not permitted to download or to forward/distribute the text or part of it without the consent of the author(s) and/or copyright holder(s), unless the work is under an open content license (like Creative Commons).

The publication may also be distributed here under the terms of Article 25fa of the Dutch Copyright Act, indicated by the "Taverne" license. More information can be found on the University of Groningen website: https://www.rug.nl/library/open-access/self-archiving-pure/taverneamendment.

If you believe that this document breaches copyright please contact us providing details, and we will remove access to the work immediately and investigate your claim.

Downloaded from the University of Groningen/UMCG research database (Pure): http://www.rug.nl/research/portal. For technical reasons the number of authors shown on this cover page is limited to 10 maximum.



university of groningen

faculty of economics and business

15005-EEF

Price Leadership and Unequal Market Sharing: Collusion in Experimental Markets

Peter T. Dijkstra



 faculty of economics and business

SOM is the research institute of the Faculty of Economics & Business at the University of Groningen. SOM has six programmes:

- Economics, Econometrics and Finance
- Global Economics & Management
- Human Resource Management & Organizational Behaviour
- Innovation & Organization
- Marketing
- Operations Management & Operations Research

Research Institute SOM Faculty of Economics & Business University of Groningen

Visiting address: Nettelbosje 2 9747 AE Groningen The Netherlands

Postal address: P.O. Box 800 9700 AV Groningen The Netherlands

T +31 50 363 7068/3815

www.rug.nl/feb/research

university of groningen

Price Leadership and Unequal Market Sharing: Collusion in Experimental Markets

Peter T. Dijkstra Faculty of Economics and Business, Department of Economics, Econometrics and Finance

Price Leadership and Unequal Market Sharing: Collusion in Experimental Markets

Peter T. Dijkstra*

April 17, 2015

Abstract

We consider experimental markets of repeated homogeneous pricesetting duopolies. We investigate the effect on collusion of sequential versus simultaneous price setting. We also examine the effect on collusion of changes in the size of each subject's market share in case both subjects set the same price. Our results show that sequential price setting compared with simultaneous price setting facilitates collusion, if subjects have equal market shares or if the follower has the larger market share. With sequential price setting, we find more collusion if subjects have equal market shares rather than unequal market shares. We observe more collusion if the follower has the larger market share than if the follower has the smaller market share.

JEL Classification Codes: C73, C92, L13, L41.

Keywords: Collusion; Price Leadership; Asymmetries; Experiment.

1 Introduction

In about one third of the cartel cases prosecuted by the European Commission, the market had a price leader and (several) price followers (Mouraviev and Rey, 2011). Examples include markets for fittings, professional videotape and candle wax (DG Competition, 2006, 2007, 2008). In a recent

^{*}e-mail: p.t.dijkstra@rug.nl. Phone: +31 50 363 4001. Fax: +31 50 363 7337. Department of Economics, Econometrics and Finance, University of Groningen, PO Box 800, 9700 AV Groningen, The Netherlands. I thank Raffaele Fiocco, Aurora García Gallego, Marco Haan, Joe Harrington, Jose Luis Moraga-González, Hans-Theo Normann, Lambert Schoonbeek, Adriaan Soetevent, and two anonymous referees for helpful comments. I am also indebted to participants of the 14th CCRP workshop (Vienna), CRESSE 2013 (Corfu), EARIE 2013 (Évora), Jornadas de Economía Industrial 2012 (Murcia) and seminar participants at the University of Groningen (RUG). Financial support of the University of Groningen (RUG) is gratefully acknowledged.

study, Mouraviev and Rey (2011) theoretically investigate the role of price leadership with regard to (tacit) collusion. They allow for the possibility of unequal market shares in case firms set the same price. They argue that sequential price setting, compared with simultaneous price setting, facilitates collusion by making it easier to punish deviations by the leader, which relaxes the incentive of the leader to deviate. Furthermore, they show that, with sequential price setting, collusion is facilitated if the follower's market share is higher, in case both firms set the same price. In particular, considering a repeated duopoly model with homogeneous goods and sequential price setting, Mouraviev and Rey demonstrate that collusion can be sustained for any discount factor, if the follower's market share is large enough. In contrast, with simultaneous price setting, collusion can only arise in equilibrium if the discount factor is large enough (Friedman, 1971).

Inspired by Mouraviev and Rey (2011), we consider experimental markets of repeated homogeneous price-setting duopolies. We investigate the effect on collusion of sequential versus simultaneous price setting. Further, we examine the effect on collusion of changes in the size of each subject's market share in case both subjects set the same price. In particular, we address the following two questions. First, does price leadership facilitate collusion, for a given type of market sharing? Second, with sequential price setting, does a larger market share of the follower facilitate collusion? There is one related issue which we will also examine. Mouraviev and Rey (2011) argue that market-share inequality in case firms set the same price, might facilitate collusion with sequential price setting. But we know from standard theory that with simultaneous price setting, market-share inequality hinders collusion (Ivaldi et al., 2003; Motta, 2004, pp. 164–165). We also investigate this claim in our experiment.

In our experiment we impose whether subjects set prices simultaneously or sequentially. We exogenously impose market shares in case subjects set the same price, to isolate the effect of market-share inequality. With simultaneous price setting we consider two treatments which differ in how the market is shared in case subjects set the same price. In one treatment the market is shared equally, in the other unequally. With sequential price setting we consider three treatments which differ in how the market is shared in case subjects set the same price. In one treatment the market is shared equally, in one the follower obtains the larger market share, and in one the follower obtains the smaller market share. To the best of our knowledge, we are the first to conduct an experiment on price leadership with homogeneous Bertrand competition, and our experiment is the first with unequal market sharing in case subjects set the same price.

Concerning the theory, there are two possible caveats in relation to our experiment. First, since we focus on tacit collusion, subjects might find it difficult to coordinate on the same collusive price. In the theoretical analyses of collusion by Motta (2004) and Mouraviev and Rey (2011), this coordination problem does not play a role. In practice, however, the coordination problem might be relevant, in particular in the case of simultaneous price setting. Scherer and Ross (1990, pp. 346–347) mention that one reason to introduce leadership in a market is indeed to facilitate tacit coordination on the same collusive price. This coordination problem is due to the unobservability of the competitor's price when one has to set her own price. A subject who wants to coordinate on the same price or wants to undercut her competitor, can therefore not be certain what her optimal strategy should be. Second, the larger the market share of the follower in case firms set the same price, the larger the difference between the collusive payoffs of the follower and leader. In the theoretical analysis of Mouraviev and Rey (2011), the utility of one firm does not depend on the profit of the other firm, but it might have an effect in practice. We know from the experimental literature that fairness arguments, in the sense that subjects dislike (large) payoff differences, matter (e.g. in the ultimatum game, see Roth, 1995; Camerer, 2003). In addition, Gibbons and Murphy (1990) and Albuquerque (2009) empirically show that CEOs of firms do not only care about their absolute performance, but also care about their relative performance. If such an effect is relevant in our experiment, it might imply that subjects are not willing to collude, if that would lead to too large payoff differences. Thus, an increase in payoff differences might (partially) offset the procollusive effect identified by Mouraviev and Rey.

In evaluating the results, we use three measures of collusion. We find the following with regard to our two main questions. First, we find more collusion with sequential price setting than with simultaneous price setting, if firms have equal market shares. We also find more collusion with sequential price setting than with simultaneous price setting if the follower has the larger market share. However, if the follower has the smaller market share, evidence is mixed. We argue that this can be explained in terms of the coordination problem and fairness arguments mentioned above. Second, with sequential price setting, a larger market share of the follower sometimes facilitates collusion. If we compare the case where the follower has the smaller market share with the cases where the market share of the follower is equal to or larger than the market share of the leader, we find more collusion in the latter cases. However, if we compare the case where the market shares of the follower and leader are equal with the case where the follower has the larger market share, we find less collusion in the latter case. Based on these results, we also compare sequential price setting and a larger follower's market share with simultaneous price setting and equal market shares, which combines a change in the type of price setting and a change in the size of market shares. We find more collusion in the former case.

The remainder of this paper is organized as follows. Section 2 provides an overview of the related literature. Section 3 presents the theoretical model

and Section 4 the experimental design. Section 5 gives our results, while Section 6 concludes. Our regression model and detailed information on the experiment are provided in the Appendices.

2 Related Literature

Our experiment considers price leadership and a type of asymmetric market sharing. In this section we discuss the literature about (i) price and quantity leadership, (ii) different types of market sharing, and (iii) different types of asymmetries between firms.

Several experimental papers discuss price and quantity leadership. Hildenbrand (2010) provides an extensive overview. Kübler and Müller (2002) compare sequential price setting with simultaneous price setting in a repeated duopoly with heterogeneous products and symmetric firms. They find that sequential price setting yields more collusion than simultaneous price setting. Huck et al. (2001) conduct an experiment on quantity leadership in homogeneous duopoly markets. They observe in a repeated game higher levels of output, and thus less collusion, with sequential quantity setting than with simultaneous quantity setting. These two experiments impose one of the subjects to take the role of the leader, while the other subject acts as the follower. In our experiment we do the same. Some other experiments consider endogenous timing where each round consists of two stages and subjects are allowed to choose in which stage to set their price or quantity. For example, Datta Mago and Dechenaux (2009) investigate repeated price-setting homogeneous duopolies with capacity-constrained firms. They find more collusion when it turns out that subjects have chosen to set prices in different stages rather than in the same stage. Furthermore, this effect is stronger with asymmetric capacity constraints. Further, Fonseca et al. (2006) consider repeated homogeneous duopolies where subjects announce when to set their quantity. They find no effect on collusion between these

two cases. Other related experiments are Huck et al. (2002) and Fonseca et al. (2005). In summary, we see that sequential price setting facilitates collusion, while sequential quantity setting hinders collusion. With endogenous timing, there is more collusion when it turns out that subjects have chosen to set prices in different stages rather than in the same stage, while there is no such effect for quantity setting.

Some experiments consider different types of market sharing. Puzzello (2008) investigates the effect on collusion of two different tie-breaking rules in case firms set the same price in a homogeneous duopoly with simultaneous price setting and capacity-constrained firms. She considers a *share* tie-breaking rule where the market is shared equally, and a *random* tie-breaking rule where each firm is selected with the same probability to supply the market first. The random tie-breaking rule implies unequal market sharing ex post, but ex ante payoffs do not differ between the two rules. Puzzello finds more collusion with the share tie-breaking rule than with the random tie-breaking rule. On the other hand, Davis and Wilson (2002) find no difference in collusion between these two tie-breaking rules in an auction with homogeneous goods and four capacity-constrained sellers per market. Thus, evidence on the effect on collusion of asymmetric market sharing is mixed. In our experiment we impose unequal market shares in a number of treatments. However, our firms are not capacity constrained.

A number of experiments focus on different types of asymmetries between firms. Mason et al. (1992) investigate quantity-setting duopolies with a homogeneous good. Each firm has either low or high constant marginal cost. The authors find more collusion if firms have equal marginal costs than if firms have different marginal costs. Dugar and Mitra (2009) vary the size of the marginal cost asymmetry in homogeneous Bertrand-duopolies under fixed matching of subjects and random assignment of marginal costs in every round. They find more collusion if the difference between the two possible values of marginal costs is smaller.¹ Phillips et al. (2011) examine heterogeneous quantity-setting duopolies and different marginal costs. They find more collusion if firms have equal marginal costs than if firms have different marginal costs. They find no difference in collusion if the difference between the possible values of marginal costs is smaller. Argenton and Müller (2012) investigate the effect of firms with different (convex) cost structures in price-setting duopoly markets with homogeneous goods. They find no difference in collusion if firms have the same cost structure or different ones. Finally, Fonseca and Normann (2008) analyze duopolies and triopolies with price competition and homogeneous goods. Firms have symmetric or asymmetric capacity constraints. Holding the number of firms constant, they find more collusion with equal capacity constraints. In summary, we see that asymmetric costs or asymmetric capacity constraints in general hinder collusion. In our experiment firms have no capacity constraints, no fixed costs and marginal costs are normalized to zero.

3 The Model

In this section we discuss the theoretical model behind our experiment. In Section 3.1 we discuss simultaneous price setting and in Section 3.2 sequential price setting. We present our hypotheses in Section 3.3.

3.1 Simultaneous Price Setting

First, consider the one-shot game with simultaneous price setting. In case both firms set the same price, firm $i \in \{1, 2\}$ receives a given share $\alpha_i \in (0, 1)$ of the aggregate profit, where $\alpha_1 + \alpha_2 = 1$. In the one-shot Bertrand-Nash equilibrium both firms set a price $p^N = 3$, i.e. the lowest possible price. We refer to this as the "competitive equilibrium" and "competitive price", respectively. Since we have inelastic unit demand and zero costs, the

¹This result is confirmed by Dugar and Mitra (2013) in the same experimental setup with random matching of subjects and fixed marginal costs.

corresponding aggregate competitive profit is given by

$$\pi^N \equiv p^N = 3. \tag{1}$$

Firm *i* receives a profit $\alpha_i p^N$ in the competitive equilibrium.

Next, suppose that both firms set a collusive price $p^C \in \{4, \ldots, 12\}$, which is larger than the competitive price.² The corresponding aggregate collusive profit is given by

$$\pi^C \equiv p^C. \tag{2}$$

Firm *i* receives profit $\alpha_i p^C$ if firms collude. In the one-shot game, a collusive price cannot be sustained as an equilibrium.

We now turn our attention to the (infinitely) repeated game. We assume that firms use grim trigger strategies (Friedman, 1971). Then, each firm will set a collusive price p^{C} in every round as long as no firm has deviated from this price. After a deviation, firms revert to the one-shot Bertrand-Nash equilibrium forever. More formally, we define $V^{C}(\alpha_{i})$ as firm *i*'s value in case both firms set a collusive price p^{C} in each round, i.e.

$$V^C(\alpha_i) \equiv \frac{\alpha_i \pi^C}{1 - \delta}.$$
(3)

Consider a unilateral deviation, and suppose that the optimal deviation yields deviation profit π^D . Note that by deviating, the firm will supply the whole market and, therefore, receive all profit in that round. However, in future rounds firm *i* receives its share α_i of the aggregate competitive profit (1). Therefore, the value of a unilateral deviation for firm *i* is

$$V^{D}(\alpha_{i}) \equiv \pi^{D} + \frac{\delta}{1-\delta} \alpha_{i} \pi^{N}.$$
(4)

Firm *i* will not deviate if and only if $V^{C}(\alpha_{i}) \geq V^{D}(\alpha_{i})$, i.e.

$$\frac{\alpha_i \pi^C}{1-\delta} \ge \pi^D + \frac{\delta}{1-\delta} \alpha_i \pi^N,\tag{5}$$

²Note that all prices above the competitive price are collusive, but the collusive price $p^{C} = 12$ Pareto dominates all other collusive prices.

which results in the critical discount factor

$$\delta \ge \delta_{sim} \left(\alpha_i \right) \equiv \frac{\pi^D - \alpha_i \pi^C}{\pi^D - \alpha_i \pi^N}.$$
(6)

An equilibrium that sustains collusion exists if and only if both firms are willing to collude, i.e. if and only if $\delta \geq \max \{\delta_{sim}(\alpha_1), \delta_{sim}(\alpha_2)\}$. From (6) it follows that $\frac{\partial \delta_{sim}}{\partial \alpha_i} < 0$, and therefore collusion is sustainable with simultaneous price setting if and only if

$$\delta \ge \delta_{sim} \Big(\min \left\{ \alpha_1, \alpha_2 \right\} \Big). \tag{7}$$

Thus, the discount factor of the smallest firm is most stringent (cf. Motta, 2004, pp. 164–165), i.e. an increase in market-share inequality makes collusion less stable.

3.2 Sequential Price Setting

Next, take the model with sequential price setting. Considering the one-shot game, in case both firms set the same price, the leader obtains a given share $\alpha_L \in (0, 1)$ of the aggregate profit and the follower's share is $\alpha_F = 1 - \alpha_L$. In the one-shot Bertrand-Nash equilibrium both firms set a price $p^N = 3$. In case both firms set the competitive price, the leader receives the competitive profit $\alpha_L p^N$, and the follower the competitive profit $\alpha_F p^N$. Similarly, if firms collude on the same price $p^C \in \{4, 5, \ldots, 12\}$, the leader receives the collusive profit $\alpha_L p^C$, and the follower the collusive profit $\alpha_F p^C$. In the one-shot game, a collusive price cannot be sustained as an equilibrium since any collusive price by the leader will be undercut in the same round by the follower and, therefore, the leader will set the competitive price.

Next, we examine the (infinitely) repeated game and assume again that firms use grim trigger strategies. We find that the leader's value of colluding is

$$V^C(\alpha_L) = \frac{\alpha_L \pi^C}{1 - \delta},\tag{8}$$

whereas the follower's value of colluding is

$$V^C(\alpha_F) = \frac{\alpha_F \pi^C}{1 - \delta}.$$
(9)

Consider a unilateral deviation by the leader. If the leader deviates, this is immediately noticed and punished by the follower. The follower undercuts the leader's deviation in the same round, implying that the leader will not obtain any profit in that round. Consecutively, in future rounds the leader receives its share α_L of the aggregate competitive profit. A deviation by the leader is never profitable, since

$$\frac{1}{1-\delta}\alpha_L \pi^C \ge \frac{\delta}{1-\delta}\alpha_L \pi^N.$$
(10)

The leader thus prefers to collude since its incentive compatibility constraint is always satisfied.

Next, consider a unilateral deviation by the follower. The follower's optimal deviation yields deviation profit π^D . Since firms use grim trigger strategies, a deviation by the follower will be followed by both firms reverting to the one-shot Bertrand-Nash equilibrium from the next round onward. Thus, in future rounds the follower receives its share α_F of aggregate competitive profit, and the value of a unilateral deviation of the follower is

$$V^D(\alpha_F) = \pi^D + \frac{\delta}{1-\delta} \alpha_F \pi^N.$$
(11)

With sequential price setting, an equilibrium sustaining collusion exists if and only if $V^{C}(\alpha_{F}) \geq V^{D}(\alpha_{F})$, i.e.

$$\frac{\alpha_F \pi^C}{1-\delta} \ge \pi^D + \frac{\delta}{1-\delta} \alpha_F \pi^N, \tag{12}$$

which results in the critical discount factor

$$\delta \ge \delta_{seq} \left(\alpha_F \right) \equiv \frac{\pi^D - \alpha_F \pi^C}{\pi^D - \alpha_F \pi^N}.$$
(13)

From (13), it follows that $\frac{\partial \delta_{seq}}{\partial \alpha_F} < 0$. Thus, increasing the follower's market share decreases the critical discount factor (cf. Mouraviev and Rey, 2011), and hence facilitates collusion.

3.3 Hypotheses

Proceeding, we now formalize our hypotheses in order to answer our research questions. For that we consider the critical discount factors in (6) with simultaneous price setting, and in (13) with sequential price setting. We first rank the models based on the largest critical discount factor given the timing of price setting. With simultaneous price setting, this is the discount factor of the subject with the smallest market share. With sequential price setting, it is the discount factor of the follower.

To investigate whether price leadership facilitates collusion for a given type of market sharing, consider the parameter allocation $\alpha_1 = \alpha_L$ and $\alpha_2 = \alpha_F$, where we denote $\alpha_- = \min \{\alpha_1, \alpha_2\}$. First, consider the case $\alpha_- =$ $\alpha_1 = \alpha_L < \alpha_F$. From (7) and (13), we obtain that $\delta_{sim}(\alpha_-) > \delta_{seq}(\alpha_F)$ if and only if $\alpha_{-} < \alpha_{F}$. Hence, we expect more collusion with sequential price setting than with simultaneous price setting if the follower's market share is larger than the leader's (in case they set the same price). Second, consider the case $\alpha_{-} = \alpha_{2} = \alpha_{F}$. From (7) and (13) we obtain $\delta_{sim}(\alpha_{-}) = \delta_{seq}(\alpha_{F})$ and, hence, the largest critical discount factors equal. We refine our ranking using the smallest critical discount factor given the timing of price setting. With simultaneous price setting, this is the discount factor of the subject with the larger market share. With sequential price setting, it is the discount factor of the leader. Recall that the leader always prefers to collude and, therefore, her discount factor equals zero. With simultaneous price setting, the discount factor of the subject with the larger market share is larger than zero and, therefore, larger than the leader's discount factor with sequential price setting. Hence, we also expect more collusion with sequential price setting than with simultaneous price setting if the follower's market share is smaller than the leader's (in case they set the same price). In sum, we thus find in either case that sequential price setting facilitates collusion in comparison with simultaneous price setting. We have:

Hypothesis 1 (Effect of Price Leadership). Irrespective of the allocation of market shares in case subjects set the same price, there is more collusion with sequential price setting than with simultaneous price setting.

Second, we examine whether, with sequential price setting, a larger market share of the follower facilitates collusion. In (13) we found that increasing the follower's market share decreases the critical discount factor and, hence, facilitates collusion. We thus consider:

Hypothesis 2 (Effect of Follower's Market Share). Suppose that subjects set prices sequentially. Then there is more collusion if the market share of the follower is larger (in case subjects set the same price).

Ultimately, as a benchmark, we also investigate whether, with simultaneous price setting, market-share inequality hinders collusion. In (7) we found that the discount factor of the smallest firm is most stringent, i.e. an increase in market-share inequality makes collusion less stable. Therefore, we have:

Hypothesis 3 (Effect of Unequal Market Sharing). Suppose that subjects set prices simultaneously. Then there is more collusion with marketshare equality than with market-share inequality (in case subjects set the same price).

4 Experimental Design

The experiment consists of a repeated price-setting duopoly game where subjects sell a homogeneous good. Demand is inelastic and normalized to unity. Costs are normalized to zero. Every subject participates in a number of duopolies which are called matches. Each match has the same structure. During one match a subject plays with the same competitor. Every match consists of a randomly-determined number of rounds. Following Roth and Murnighan (1978), we simulate an infinitely repeated game by implementing a given continuation probability after every round.³ With probability $\delta \in (0, 1)$, two subjects play another round with each other. With probability $1 - \delta$, the current match ends. This implies that the expected number of rounds in a match is $\frac{1}{1-\delta}$. We impose a continuation probability of $\delta = 0.70$, which is common knowledge among subjects. Each pair is therefore expected to be matched for $3\frac{1}{3}$ rounds.⁴ This matching protocol ensures that we do not have to be concerned about possible end-game effects, as every round played is potentially the last one. It also provides subjects the opportunity to try different strategies over time as every new match is a new beginning.

We run five treatments which differ in two dimensions: subjects set prices simultaneously or sequentially in each round; and the market is shared equally or unequally in case subjects set the same price. Our treatments with simultaneous price setting consist of two stages in each round. In stage 1, subjects choose their prices simultaneously and independently. In stage 2, subjects learn the price chosen by their competitor and profits are realized. Our treatments with sequential price setting consist of three stages in each round. In stage 1, the leader chooses her price. In stage 2, the follower learns the price chosen by the leader. Subsequently, the follower chooses her own price. In stage 3, the leader learns the price chosen by its competitor and profits are realized. In either setup, a subject learns the price chosen by her competitor, and the resulting profits for herself and her competitor.

In every round, subjects choose a price from the set $\{3, 4, ..., 12\}$. The prices 1 and 2 are excluded to ensure uniqueness of the equilibrium in the one-shot game (see also Dufwenberg and Gneezy, 2000). If a subject sets the lowest price, she⁵ captures the entire market and makes profit equal to her price. If subjects set the same price, the division of the market

 $^{^{3}}$ This setup is also implemented in the experiments of, amongst others, Dal Bó (2005), Blonski et al. (2011), Dal Bó and Fréchette (2011), Bigoni et al. (2012) and Cason et al. (2013).

 $^{^4\}mathrm{Appendix}$ B provides the actual number of rounds played in each match in each session. $^5\mathrm{We}$ refer to a subject as "she".

depends on the treatment. To isolate the effect of market-share inequality and to simplify the experiment, we exogenously impose market shares in case subjects set the same price.⁶

In SIMEQUAL we have simultaneous price setting and each subject obtains a share of 50% of the market in case subjects set the same price. In all other treatments we have two different types of players, A and B. Half of the subjects were randomly assigned to role A and the other half to role B. Subjects kept their role throughout the session. In each match an A-player was matched with a B-player. In the other treatment with simultaneous price setting, SIMUNEQUAL, subject A obtains a share of 30% of the market while subject B obtains a share of 70%, in case both players charge the same price.⁷ In the treatments with sequential price setting, subject A is the price leader while subject B is the price follower. We have three treatments with sequential price setting which differ in how the market is shared in case subjects set the same price. In FOLLOWER30 the follower obtains a share of 30% of the market while the leader obtains a share of 70%. In FOLLOWER50 each subject obtains a share of 50% of the market. In FOLLOWER70 the follower obtains a share of 70% of the market while the leader obtains a share of 30%.

Information on the reasons of implementing unequal market shares is not provided to the subjects in our experiment. However, they were presented the structure of profits and a payoff table. The treatments are summarized

⁶This is a difference between our experimental setup and Mouraviev and Rey's (2011) theoretical model: they assume that firms can share the market as they wish in case they charge the same price.

⁷From the ultimatum game literature it is known that payoff differences matter (Roth, 1995; Camerer, 2003). In the ultimatum game, one of two subjects proposes a division of a fixed amount of money. The other subject, the responder, either accepts or rejects this proposal. If she accepts, the money is divided according to the proposal. If the responder rejects, each receives nothing. Oosterbeek et al. (2004) find in their meta analysis of the ultimatum game that the probability of acceptance increases in the percentage of money offered to the responder. Furthermore, most offers below 20% are not accepted. We presume that in our experiment the smallest share should be above 20%, but also not too close to 50%. We decided to set the smallest market share at 30%.

Table 1: Treatment characteristics.

Treatment	Price setting	Market shares in case subjects set the same price				
			Subject A	Subject B		
SimEqual	Simultanoous	Equal	$\alpha_1 = 50\%$	$\alpha_2 = 50\%$		
SimUnequal	Simultaneous	Unequal	$\alpha_1 = 30\%$	$\alpha_2 = 70\%$		
Follower30		Unequal	$\alpha_L = 70\%$	$\alpha_F = 30\%$		
Follower50	Sequential	Equal	$\alpha_L = 50\%$	$\alpha_F = 50\%$		
Follower70		Unequal	$\alpha_L = 30\%$	$\alpha_F = 70\%$		

In the treatments with sequential price setting subject A is the price leader while subject B is the follower.

in Table 1.

Once a match ends, subjects are matched to create new duopolies. All treatments, except SIMEQUAL, used an absolute type stranger design, i.e. each A-player was matched exactly once to each B-player and vice versa. Therefore, the total number of matches in a session is $\frac{P}{2}$, where P is the number of subjects in a session. Since we had 16 or 18 subjects in each session, 8 or 9 matches were being played. In SIMEQUAL an absolute stranger design was used. Because there is a maximum of 9 matches in the other treatments, we randomly matched each subject in SIMEQUAL to 9 unique other subjects. Therefore, all sessions had the same expected total number of rounds.

5 Results

The experiment has been conducted at the Groningen Experimental Economics Laboratory (GrEELab) at the University of Groningen in 2012. A total of 176 subjects participated which were all students at the Faculty of Economics and Business (98.3%) or at other faculties of that university (1.7%). Every session consisted of one treatment and lasted between 60 and 90 minutes. Every treatment was run twice. Treatments were randomly assigned to sessions, and either 16 or 18 subjects participated in a session.

The experiment was programmed in z-Tree (Fischbacher, 2007). Printed

instructions were provided and read aloud.⁸ Subjects first had to answer a number of questions correctly on their computer to ensure understanding of the experiment.

Subjects were paid their cumulative earnings in euros at a rate of ≤ 0.07 per point, including an initial endowment of 75 points. Average earnings were ≤ 11.49 and ranged from ≤ 7.30 to ≤ 24.60 . Detailed information on each session, including the number of rounds played in each match, and average, minimum and maximum earnings, is provided in Appendix B.

5.1 Measures of Collusion

We measure collusion in three different ways. First, we consider the **incidence of supra-competitive prices**, i.e. the percentage of markets where, in a given round, the actual market price exceeds the competitive price $p^N = 3$. Figure 1 shows the incidence of supra-competitive prices over time for each treatment.⁹ All treatments incur many supra-competitive prices in the first round. After that the incidence decreases significantly over time for all five treatments (see Table 2).¹⁰

Second, we consider the **incidence of price coordination**. This is the percentage of markets where, in a given round, both subjects charge the same collusive price $p^C \in \{4, 5, ..., 12\}$.¹¹ This is a stricter measure than the incidence of supra-competitive prices. Figure 2 shows the incidence of price coordination over time for each treatment. In most treatments the incidence of price coordination in each round is much lower than the inci-

⁸Appendix D reproduces instructions for FOLLOWER70. Instructions for other treatments are similar and available upon request.

⁹Averages are calculated per round over all active groups. Note that the number of active groups is not constant over rounds.

 $^{^{10}}$ This has also been found in different contexts by Mason et al. (1992), Dal Bó (2005), Dugar and Mitra (2009), Blonski et al. (2011), and Dugar and Mitra (2013) who also found that cooperation decreases over time.

¹¹There are a few markets where subjects manage to take turns in supplying the market by alternating in prices. Results remain qualitatively the same when we include price alternation in our definition of price coordination. We also obtain qualitatively similar results if we focus on coordination on the highest collusive price of 12, which is a stricter measure of price coordination.



Figure 1: Incidence of supra-competitive prices per round (average across all active groups).

Table 2: Time trend regressions for incidence of supra-competitive prices per treatment.

	C:	C:1	F-1120	E-11F0	F-1170
	SimEqual	SimUnequal	Follower30	Follower50	Follower70
Round	-0.171**	-0.506***	-0.286***	-0.0596+	-0.112^{*}
	(0.0647)	(0.129)	(0.0647)	(0.0348)	(0.0439)
Constant	0.925***	1.518***	0.571**	1.076***	1.025***
	(0.171)	(0.268)	(0.183)	(0.158)	(0.160)
Observations	486	531	567	691	544
Pseudo R^2	0.027	0.130	0.052	0.011	0.025

Clustered standard errors on group level in parentheses. $^+$ denotes significance at the 10% level; * at 5%; ** at 1%; *** at 0.1%.



Figure 2: Incidence of price coordination per round (average across all active groups).

dence of supra-competitive prices, but in FOLLOWER50 they are close to each other. Hence, in that case, supra-competitive prices are frequently accompanied by price matching. From the figure and Table 3 we see that in SIMEQUAL, FOLLOWER50 and FOLLOWER70 the incidence of price coordination is roughly stable over time, while there is a decreasing time trend in SIMUNEQUAL and FOLLOWER30.

Third, we investigate the magnitude of **market prices**. Figure 3 shows the average price path for each treatment. Market prices in the treatments with sequential price setting are almost always higher than in the treatments with simultaneous price setting. Again, there is a decreasing time trend in SIMUNEQUAL and FOLLOWER30 as confirmed in Table 4.

We will next discuss the results for each hypothesis.¹² Comparing the

 $^{^{12}}$ We include data from all matches and all rounds. In a few sessions we find significant learning of subjects over the first matches, and we observe within matches some negative time trends in the data (results are available from the author upon request). In order to correct for the existence of these learning effects, we also considered our results if we

	SimEqual	SimUnequal	Follower30	Follower50	Follower70
Round	-0.0289	-0.288*	-0.204*	-0.0310	-0.0613
	(0.0950)	(0.120)	(0.0815)	(0.0323)	(0.0556)
Constant	-1.615***	-1.155***	-1.209***	0.329 +	-0.690***
	(0.230)	(0.280)	(0.253)	(0.172)	(0.208)
Observations	486	531	567	691	544
Pseudo R^2	0.001	0.033	0.021	0.003	0.006

Table 3: Time trend regressions for incidence of price coordination per treatment.

Clustered standard errors on group level in parentheses. $^+$ denotes significance at the 10% level; * at 5%; ** at 1%; *** at 0.1%.



Figure 3: Market price per round (average across all active groups).

	SimEqual	SimUnequal	Follower30	Follower50	Follower70
Round	-0.00768	-0.238***	-0.251**	-0.0226	-0.0977
	(0.0924)	(0.0325)	(0.0754)	(0.0701)	(0.0735)
Constant	4.544***	4.755***	5.931***	7.432***	6.411***
	(0.157)	(0.102)	(0.278)	(0.318)	(0.259)
Observations	486	531	567	691	544
R^2	0.000	0.124	0.026	0.001	0.010
Adjusted R^2	-0.002	0.122	0.025	-0.001	0.008

Table 4: Time trend regressions for market price per treatment.

Clustered standard errors on group level in parentheses. $^+$ denotes significance at the 10% level; * at 5%; ** at 1%; *** at 0.1%.

relevant treatments in a pairwise fashion, all statistical tests reported below are for the no-treatment effect versus the two-sided alternative, as outlined in Appendix A.

5.2 Effect of Price Leadership

First, we examine Hypothesis 1. We begin with the scenario where subjects have equal market shares in case they set the same price. Hypothesis 1 then implies more collusion in FOLLOWER50 than in SIMEQUAL. Table 5 reports on these treatments. We obtain the following result.

Result 1a (Effect of Price Leadership). There is more collusion in FOLLOWER50 than in SIMEQUAL, when measured by the incidence of price coordination or the level of market prices.

Hence, for two out of three measures of collusion we find support for Hypothesis 1 when subjects have equal market shares (in case they set the same price). The incidence of supra-competitive prices is also higher in FOLLOWER50 than in SIMEQUAL, but there the difference is not significant.

exclude the first 1/2/3 match(es); if we exclude the first and last match; or if we exclude rounds 10 and further. Our results remain qualitatively the same in all cases. The only difference is the comparison of SIMUNEQUAL and FOLLOWER30 in Table 6 where the incidence of supra-competitive prices would become insignificant if we exclude the first two or three matches.

	SimEqual		Follower50
Supra-competitive Prices	59.9%	\approx	68.5%
	(2.2%)		(1.8%)
Price Coordination	15.4%	<***	54.4%
	(1.6%)		(1.9%)
Market Price	4.52	<***	7.32
	(0.08)		(0.14)

Table 5: Comparison of SIMEQUAL and FOLLOWER50 (across all rounds and active groups).

Standard errors in parentheses. Entries between values indicate whether the value to the left is significantly lower (<), significantly higher (>), or does not differ significantly (\approx) from the value to the right. Differences between treatments are tested using regressions with clustered standard errors on group level as outlined in Appendix A. *** denotes significance at the 0.1% level.

Next, consider the scenario where subjects have unequal market shares in case they set the same price. Hypothesis 1 then implies more collusion in both FOLLOWER30 and FOLLOWER70 than in SIMUNEQUAL. Table 6 reports on these treatments. We have the following result.

Result 1b (Effect of Price Leadership).

- a. There is more collusion in FOLLOWER70 than in SIMUNEQUAL, for all three measures of collusion.
- b. There is more collusion in FOLLOWER30 than in SIMUNEQUAL, when measured by the level of market prices. However, there is more collusion in SIMUNEQUAL than in FOLLOWER30, when measured by the incidence of supra-competitive prices.

Hence, when the follower has the larger market share we find strong support for Hypothesis 1. However, when the follower has the smaller market share, evidence is mixed. We find support for Hypothesis 1 using market prices as a measure of collusion, but the hypothesis is rejected using supra-competitive prices as a measure.

	Follower30		SimUnequal		Follower70
Supra-competitive Prices	42.7%	<*	52.4%	<*	64.3%
	(2.1%)		(2.2%)		(2.1%)
Price Coordination	14.3%	\approx	13.2%	<***	28.7%
	(1.5%)		(1.5%)		(1.9%)
Market Price	5.14	>***	4.07	<***	6.04
	(0.13)		(0.06)		(0.14)

Table 6: Comparison of SIMUNEQUAL with FOLLOWER30 and FOLLOWER70 (across all rounds and active groups).

Standard errors in parentheses. Entries between values indicate whether the value to the left is significantly lower (<), significantly higher (>), or does not differ significantly (\approx) from the value to the right. Differences between treatments are tested using regressions with clustered standard errors on group level as outlined in Appendix A. * denotes significance at the 5% level; *** at 0.1%.

A possible explanation for this mixed result is the following. We distinguish two different effects. First, consider the subject with a market share of 30% in case subjects set the same price. In FOLLOWER30, a follower who wants to undercut the leader can simply set a price that is one unit lower than the leader's price.¹³ In SIMUNEQUAL, a subject who aims to undercut her competitor, cannot observe the other's price. Therefore, she will be more careful and generally set a somewhat lower price than her counterpart in FOLLOWER30. This implies an upward pressure on prices in FOLLOWER30 vis-à-vis SIMUNEQUAL.¹⁴ It does not affect the incidence of supra-competitive prices in either treatment, as long as subjects still set a price above 3.

Second, consider the subject with a market share of 70% in case subjects

 $^{^{13}}$ The leader sets a collusive price in 46.7% of the cases in FOLLOWER30. Subsequently, the follower matches the leader's price in just 30.6% of these cases while the follower undercuts in 64.9% of these cases. More information can be found in Table 9.

 $^{^{14}}$ In SIMUNEQUAL 64.6% of the subjects with the smaller market share and 63.5% of the subjects with the larger market share set a collusive price. In those cases, the prices 4–7 are chosen by 84.0% of the subjects with the smaller market share and by 85.5% of the subjects with the larger market share. When a collusive price is chosen in FOLLOWER30, a price in this range is chosen by 49.8% of the leaders and by 41.8% of the followers. Appendix C provides a complete overview of the prices chosen by every type of subject in each treatment.

set the same price. In FOLLOWER30 it turns out that many leaders set a price equal to 3.¹⁵ Presumably, these leaders reason that if they would set a price larger than 3, the follower would not be willing to match this price and thus accept only 30% of the corresponding collusive profit while the leader would obtain the remaining 70%. Hence, these leaders anticipate the followers to use a fairness argument in the sense that followers dislike outcomes where they receive (much) less than the leader, and instead prefer to undercut the leader's price.¹⁶ This fairness argument is related to the finding in the ultimatum game literature that many subjects dislike payoff differences (Roth, 1995; Camerer, 2003), and in particular dislike it when they receive less than others (Fehr and Schmidt, 1999). In SIMUNEQUAL the fairness argument is less pervasive because it is more uncertain who will supply the market. This is due to the unobservability of the competitor's price when one has to set her own price. If the subject with a market share of 70% would draw her price from the same distribution as the subject with a market share of 30%, then in expectation the subjects would share the market equally and hence fairness arguments do not play a role. This is indeed what we observe in our experiment.¹⁷ Hence, in SIMUNEQUAL fairness is less of an issue than in FOLLOWER30. This implies a downward pressure on both prices and the incidence of supra-competitive prices in FOLLOWER30 vis-à-vis SIMUNEQUAL.

 $^{^{15}}$ In FOLLOWER30 the leader set the minimum price in 53.3% of the cases. In SIMUNEQUAL 35.4% of the subjects with the smaller market share and 36.5% of the subjects with the larger market share set a price equal to 3.

¹⁶ We did not estimate subjects' preferences for fairness in our experiment, because it might interfere with the experiment itself. If we would ask subjects about their fairness preferences before the experiment starts, subjects' attention might be drawn to the unequal division of profits in case they set the same price. Hence, that could trigger some subjects to look for an equal division of profits during the experiment. On the other hand, asking subjects about their fairness preferences after the experiment would probably lead to an overstatement of their preferences for equality in the treatments with an asymmetric tie-breaking rule if they earned (much) less than their competitors.

¹⁷In SIMUNEQUAL, a Kolmogorov-Smirnov test for equality of distribution functions indicates that the distribution of prices chosen by the subject with the smaller market share is not significantly different from the distribution of prices chosen by the subject with the larger market share (p-value = 0.999).

	Follower30	Follower50	Follower70	Follower30
Supra-competitive Prices	42.7% <**	** $68.5\% \approx$	64.3% >***	* 42.7%
	(2.1%)	(1.8%)	(2.1%)	(2.1%)
Price Coordination	14.3% <**	** $54.4\% > ***$	* 28.7% >***	* 14.3%
	(1.5%)	(1.9%)	(1.9%)	(1.5%)
Market Price	5.14 <*'	** $7.32 > **$	$6.04 >^{**}$	5.14
	(0.13)	(0.14)	(0.14)	(0.13)

Table 7: Comparison of FOLLOWER30, FOLLOWER50 and FOLLOWER70 (across all rounds and active groups).

Standard errors in parentheses. Entries between values indicate whether the value to the left is significantly lower (<), significantly higher (>), or does not differ significantly (\approx) from the value to the right. Differences between treatments are tested using regressions with clustered standard errors on group level as outlined in Appendix A. ** denotes significance at the 1% level; *** at 0.1%.

Combining the two arguments above, we obtain an unambiguously negative effect on the incidence of supra-competitive prices in FOLLOWER30 in comparison with SIMUNEQUAL, which is confirmed by our results. However, the two arguments imply countervailing effects on the size of the market price. We find the first upward effect to be dominating.

The absence of a fairness argument in SIMUNEQUAL also explains why our results support Hypothesis 1 when we compare SIMUNEQUAL with FOL-LOWER70. In FOLLOWER70 the follower obtains a share of 70% of the collusive profit if she matches the leader's price. Leaders therefore anticipate that the followers are willing to match a high collusive price set by them.

5.3 Effect of Follower's Market Share

Second, we examine Hypothesis 2. It implies more collusion in FOLLOWER70 than in both FOLLOWER50 and FOLLOWER30, and more collusion in FOLLOWER50 than in FOLLOWER30. Table 7 reports on these treatments. Note that FOLLOWER30 is listed twice in this table, to facilitate all pairwise comparisons. We obtain the following result.

Result 2 (Effect of Follower's Market Share).

Suppose that subjects set prices sequentially.

- a. There is more collusion in both FOLLOWER70 and FOLLOWER50 than in FOLLOWER30, for all three measures of collusion.
- b. There is less collusion in FOLLOWER70 than in FOLLOWER50, when measured by the incidence of price coordination or the level of market prices.

Hence, we find that the effect on collusion of a larger market share of the follower is an inverted u-shape: we find more collusion if subjects have equal market shares rather than unequal market shares, but we observe more collusion if the follower has the larger market share than if the follower has the smaller market share. We thus find strong support for Hypothesis 2 when the follower's market share changes from 30% to 50% or from 30% to 70%. However, when the follower's market share changes from 50% to 70%, we reject Hypothesis 2 for both the incidence of price coordination and the level of market prices as measures of collusion.¹⁸ The incidence of supracompetitive prices is also higher in FOLLOWER50 than in FOLLOWER70, but the difference is not significant.

The result of FOLLOWER50 versus FOLLOWER70 can be understood as follows. Figure 4 shows the price chosen by the leader in all rounds per treatment with sequential price setting.¹⁹ For each treatment, the size of each vertical bar shows which percentage of leaders has set the corresponding price. Within each bar we indicate, respectively, which percentage of the followers has set a lower price than the leader, has matched the leader's price, or has set a higher price than the leader. We first investigate the level of market prices. Here, a fairness argument can explain why market prices

¹⁸This result is, however, in accordance with Puzzello (2008) who also found more collusion if the market is shared more equally, although theory predicted no difference between equal and unequal market sharing there.

¹⁹Appendix C provides a complete overview of the prices chosen by the leader and the follower in each treatment.

Figure 4: Distribution of leader's prices and follower's responses with sequential price setting (across all rounds and active groups).



are higher in FOLLOWER50 than in FOLLOWER70. Suppose that the follower will always match the price of the leader. In FOLLOWER50, there will then be no difference in profits between follower and leader. On the other hand, in FOLLOWER70 the difference in profit received by both subjects is 40% of the chosen price. If the leader wants the absolute difference in profit not to be too large, she can empower this by charging a lower price.²⁰ This implies a downward pressure on prices in FOLLOWER70 vis-à-vis FOLLOWER50. Table 8 shows the exact cumulative distribution of leader's price, which confirms that leaders in FOLLOWER50 tend to set higher prices more often than in the other sequential treatments. The difference in leader's average price between FOLLOWER70 and FOLLOWER50 is not significant however (p = 0.133).²¹

²⁰Note that the leader intentionally hurts himself by charging a lower price. From Fehr and Schmidt (1999) we know that subjects are willing to punish other subjects, even if this is costly for themselves.

²¹The difference between FOLLOWER30 and FOLLOWER50 is statistically significant (p = 0.011) while the difference between FOLLOWER30 and FOLLOWER70 is not (p = 0.136).

Table 8: Cumulative distribution of leader's prices with sequential price setting (across all rounds and active groups).

Treatment		Price of Leader									
	Mean	3	4	5	6	7	8	9	10	11	12
Follower30	5.45	53.3%	60.3%	67.4%	71.4%	76.5%	78.3%	79.7%	81.6%	86.2%	100%
Follower50	7.48	29.4%	35.0%	41.4%	46.9%	51.1%	54.7%	59.3%	65.1%	69.3%	100%
Follower70	6.37	29.8%	40.4%	50.2%	60.8%	66.2%	72.6%	74.8%	80.3%	88.0%	100%

Second, we investigate the incidence of price coordination. Consider the follower in a given round. She might consider her current profit to be too low if she would match the price of the leader. In particular, she might even be willing to sacrifice current profit by setting a higher price than the leader, thereby showing that she is interested in higher prices in future rounds. Since (future) profits are increasing in the follower's market share, we expect this effect to be stronger in FOLLOWER70 than in FOLLOWER50. Table 9 confirms that more followers set higher prices than the leader in FOLLOWER70 than in FOLLOWER50 (p = 0.025).²² Furthermore, we also see in Table 9 that more followers undercut 23 in FOLLOWER70 than in FOLLOWER50 at every price of the leader (undercutting for all collusive prices are significantly different between these treatments, p = 0.005).²⁴ A possible explanation for this is the following, where we distinguish between two types of followers. Consider the type of follower in a given round, who has set a higher price than the leader in the previous round. Then, irrespective of whether the leader has increased her price in comparison to the previous round, a number of these followers undercut the leader²⁵ to compensate their sacrificed

²²Both pairwise comparisons with FOLLOWER30 are insignificant.

 $^{^{23}}$ A follower typically undercuts the leader by 1 unit, which is the best static response. Undercutting by more than 1 unit only happens in 2 of 172 cases in FOLLOWER30, in 4 of 97 cases in FOLLOWER50, and in 5 of 168 cases in FOLLOWER70.

²⁴We also see that even more followers undercut in FOLLOWER30 than in FOLLOWER70 at every leader's price (p = 0.039). Hence, there is more undercutting in FOLLOWER30 than in FOLLOWER50 (p = 0.000).

 $^{^{25}}$ In FOLLOWER70 57.6% of these followers under cut the leader after having set a higher price than the leader in the previous round. In FOLLOWER50 only 8.5% of the followers do this.

Price of	-	Follower3	0	-	Follower50			Follower70			
Leader	Lower	Match	Higher	Lower	Match	Higher	Lower	Match	Higher		
3		73.2%	26.8%		78.8%	21.2%		66.7%	33.3%		
4	57.5%	32.5%	10.0%	30.8%	64.1%	5.1%	53.4%	20.7%	25.9%		
5	62.5%	27.5%	10.0%	25.0%	61.4%	13.6%	41.5%	45.3%	13.2%		
6	82.6%	17.4%	0.0%	15.8%	73.7%	10.5%	43.1%	37.9%	19.0%		
7	51.7%	44.8%	3.4%	20.7%	75.9%	3.4%	48.3%	34.5%	17.2%		
8	50.0%	40.0%	10.0%	12.0%	88.0%	0.0%	42.9%	40.0%	17.1%		
9	62.5%	37.5%	0.0%	18.8%	78.1%	3.1%	41.7%	50.0%	8.3%		
10	81.8%	18.2%	0.0%	12.5%	85.0%	2.5%	36.7%	43.3%	20.0%		
11	69.2%	23.1%	7.7%	41.4%	58.6%	0.0%	50.0%	33.3%	16.7%		
12	67.9%	32.1%		17.0%	83.0%		36.9%	63.1%			
Total	30.3%	53.3%	16.4%	14.0%	77.6%	8.4%	30.9%	48.5%	20.6%		
Relevant	64.9%	30.6%	19.0%	19.9%	77.0%	12.1%	44.0%	40.8%	23.4%		

Table 9: Distribution of follower's responses with sequential price setting, by price of the leader (across all rounds and active groups).

The table provides the percentual distribution of actions subsequently taken by the follower for each price of the leader. The row "Total" provides the distribution of follower's responses for all prices of the leader. The row "Relevant" provides the percentage of follower's undercutting or matching any collusive price of the leader, and the percentage of followers setting higher prices for all prices of the leader except 12.

profits of the previous round. Next, consider another type of follower in a given round. In FOLLOWER50, the difference in follower's profit between matching and undercutting the price of the leader is 50%. In FOLLOWER70 this difference is 30% and, thus, smaller than in FOLLOWER50. Furthermore, the difference in leader's profit is smaller in FOLLOWER70 than in FOLLOWER50. This implies that, based on a fairness argument²⁶, the additional disutility of undercutting instead of matching the price of the leader is smaller in FOLLOWER70 than in FOLLOWER70 than in FOLLOWER70 than in FOLLOWER70 than in FOLLOWER50. Therefore, followers might be less reluctant to undercut the price of the leader in FOLLOWER70 than in FOLLOWER50.

Finally, it seems in Table 9 that the probability of undercutting (setting a higher price) is increasing (decreasing) in the leader's price. We formally test this by estimating regressions for each treatment in which the leader's

 $^{^{26}}$ Recall that we can only provide verbal arguments concerning fairness. See also footnote 16.

Response of Follower	Undercut	Setting same price	Setting higher price
Follower30	0.0447	-0.00155	-0.422*
	(0.0512)	(0.0568)	(0.181)
Follower50	-0.0486	0.112^{*}	-0.405**
	(0.0639)	(0.0502)	(0.133)
Follower70	-0.0429	0.125^{*}	-0.133+
	(0.0663)	(0.0526)	(0.0775)

Table 10: Influence of leader's price on follower's response per treatment.

Logistic regressions for relevant leader's prices. Standard errors in parentheses. Coefficients are reported for the influence of leader's price on the follower's response. Constants are not reported. $^+$ denotes significance at the 10% level; * at 5%; ** at 1%; *** at 0.1%.

price is the explanatory variable. The results can be found in Table 10. We see that there is no relation between the price of the leader and the level of undercutting. The probability of setting the same price as the leader is increasing in price for FOLLOWER50 and FOLLOWER70. Finally, the probability that the follower sets a higher price than the leader is decreasing in leader's price (as expected), and is significant for all treatments. The impact of the leader's price on follower's response is different between treatments, and pairwise comparison yields that the influence of leader's price on setting higher prices is higher in FOLLOWER70 than in FOLLOWER50 (p = 0.073) and the influence of leader's price on setting the same price in is higher in FOLLOWER30 (p = 0.096).

5.4 Combined Effect of Price Leadership and Follower's Market Share

Based on Hypotheses 1 and 2, we expected more collusion in FOLLOWER50 than in SIMEQUAL and more collusion in FOLLOWER70 than in FOLLOWER50, respectively. Combining this, we expect more collusion in FOLLOWER70 than in SIMEQUAL. In the experiment we found indeed more collusion in FOLLOWER50 than in SIMEQUAL. However, we found less collusion in FOL-LOWER70 than in FOLLOWER50. This raises the question whether FOL-

	SimEqual		Follower70
Supra-competitive Prices	59.9%	\approx	64.3%
	(2.2%)		(2.1%)
Price Coordination	15.4%	<**	28.7%
	(1.6%)		(1.9%)
Market Price	4.52	<***	6.04
	(0.08)		(0.14)

Table 11: Comparison of SimEqual and Follower70 (across all rounds and active groups).

Standard errors in parentheses. Entries between values indicate whether the value to the left is significantly lower (<), significantly higher (>), or does not differ significantly (\approx) from the value to the right. Differences between treatments are tested using regressions with clustered standard errors on group level as outlined in Appendix A. ** denotes significance at the 1% level; *** at 0.1%.

LOWER70 is more or less collusive than SIMEQUAL. We compare these treatments in Table 11. We obtain the following result.

Result 3 (Effect of Price Leadership and Follower's Market Share).

There is more collusion in FOLLOWER70 than in SIMEQUAL, when measured by the incidence of price coordination or the level of market prices.

Hence, for two out of three measures of collusion, we see that the procollusive effect of sequential price setting versus simultaneous price setting is larger than the procompetitive effect of the larger market share of the follower. The incidence of supra-competitive prices is also higher in FOLLOWER70 than in SIMEQUAL, but the difference is not significant.

5.5 Effect of Unequal Market Sharing with Simultaneous Price Setting

Ultimately, we examine Hypothesis 3. It implies more collusion in SIME-QUAL than in SIMUNEQUAL. Table 12 reports on these treatments. We obtain the following result.

Result 4 (Effect of Unequal Market Sharing). There is more collusion in SIMEQUAL than in SIMUNEQUAL, when measured by the level of market

	SimEqual		SimUnequal
Supra-competitive Prices	59.9%	\approx	52.4%
	(2.2%)		(2.2%)
Price Coordination	15.4%	\approx	13.2%
	(1.6%)		(1.5%)
Market Price	4.52	>*	4.07
	(0.08)		(0.06)

Table 12: Comparison of SIMEQUAL and SIMUNEQUAL (across all rounds and active groups).

Standard errors in parentheses. Entries between values indicate whether the value to the left is significantly lower (<), significantly higher (>), or does not differ significantly (\approx) from the value to the right. Differences between treatments are tested using regressions with clustered standard errors on group level as outlined in Appendix A. * denotes significance at the 5% level.

prices.

Hence, for one out of three measures of collusion we find support for Hypothesis 3. The incidence of supra-competitive prices and the incidence of price coordination are also higher in SIMEQUAL than in SIMUNEQUAL, but the differences are not significant.

6 Conclusion

In this paper we considered experimental markets of repeated homogeneous price-setting duopolies. We investigated the effect on collusion of sequential versus simultaneous price setting. We also examined the effect of changes in the size of each subject's market share in case both subjects set the same price. In particular, we addressed the following two questions.

First, does price leadership facilitate collusion, for a given type of market sharing? Our findings provide evidence that price leadership facilitates collusion, if subjects have equal market shares in case they set the same price. With unequal market shares, price leadership facilitates collusion only if the follower has the larger market share. Evidence is mixed if the follower has the smaller market share. Second, with sequential price setting, does a larger market share of the follower facilitate collusion? This is partially confirmed in our experiment. If we compare the case where the follower has the smaller market share with the cases where the market share of the follower is equal to or larger than the market share of the leader, we find more collusion in the latter cases. However, if we compare the case where the market shares of the follower has the larger market share, we find leader are equal with the case where the follower has the larger market share, we find less collusion in the latter case.

Based on these results, we also examined whether the combination of price leadership and a larger market share of the follower facilitates collusion in comparison with simultaneous price setting and equal market shares. We confirm this in our experiment. Hence, the procollusive effect of price leadership is more important here than the procompetitive effect of the increase in the market share of the follower.

Our results which contradict our expectations, might be explained in terms of the coordination problem and fairness arguments. This latter effect does not only exist in the experimental laboratory, but also exists in real markets. Gibbons and Murphy (1990) and Albuquerque (2009) empirically show that CEOs of firms do not only care about their absolute performance, but also care about their relative performance.

Based on our results, we believe that antitrust authorities should scrutinize markets with price leadership, since price leadership is a possible indication of collusion. Furthermore, markets where firms share the market equally are also more susceptible of collusion. With price leadership, there is more collusion in markets where the follower has the larger market share, than if the follower has the smaller market share.

We mention some open questions which would be interesting for further research. In our experiment we impose whether a firms is the leader or the follower. We can also impose firms to take turns in being the leader, as happened in, e.g., the Australian gasoline market (Wang, 2009). In that case, the effect on collusion of an increase in the follower's market share might be different, because the fairness argument is less strong than in our current setup. Further, we are also interested whether subjects will be able to coordinate on a collusive outcome in a setting with endogenous timing, where subjects are allowed to choose in which stage to set their price. Another open question is whether price leadership facilitates collusion if subjects can communicate. We know that communication leads to more collusion in oligopolies with simultaneous price setting is not investigated yet. Finally, it remains to be seen what the effect is on collusion if a subject's payoff depends on her own and her competitor's profit, which would also decrease the fairness argument compared with our current setup.

Appendix

A Regression Model

A.1 Group Level

Subjects had to decide which price to charge in every round. Because subjects of a group possibly interact with each other for several rounds, there might be correlation between observations of the same group. We estimate an econometric model²⁷ by adopting a regression model with clustered standard errors to account for the above-mentioned correlations, following Kübler and Müller (2002) and Dal Bó (2005).

If the variable of interest is the incidence of supra-competitive prices, then $y_{rgs} = 1$ if the market price in round $r \in \{1, 2, \ldots, R_{gs}\}$, group $g \in \{1, 2, \ldots, G_s\}$ in session $s \in \{1, 2, \ldots, S\}$ exceeds the competitive price of 3, and $y_{rgs} = 0$ otherwise. The variable y is defined similarly if we consider whether a group coordinated on the same collusive price. If we look at market prices, then y_{rgs} is the market price in round r of group g in session s. Differences between treatments are tested in a pairwise fashion. Every treatment is run twice, thus S = 4. Since we have 16 or 18 subjects in each session, each subject is matched with 8 or 9 other subjects. This implies that $G_s \in \{64, 81\}$ is the number of groups that played in session s. It turned out that every group played at most 16 rounds (see Table B.1), thus $R_{gs} \in \{1, 2, \ldots, 16\}$ is the number of rounds played by group g in session s.

We estimate the following regression model with clustered standard errors at the group level (following Kübler and Müller, 2002) to test for differences in the variable y between treatments a and b:

$$y_{rgs} = \beta_0 + \beta_1 \text{treatment}_{gs} + \epsilon_{rgs}, \tag{A.1}$$

²⁷Note that the session average of the variable of interest would be one independent observation. Thus, we would have two independent observations per treatment. The Mann-Whitney U test is only defined for at least 3 independent observations per treatment. Therefore, non-parametric tests cannot be used.

where β_0 and β_1 are coefficients to be estimated, ϵ_{rgs} are normally distributed errors, and treatment_{gs} is a dummy that equals 1 if group g in session s participated in treatment a, and 0 otherwise.

Model (A.1) allows for possible correlations between the errors ϵ_{rgs} over rounds r for a given group g in session s. Furthermore, we assume that the errors of group g and group $g' \neq g$ in session s are not correlated, and that the errors of group g in session s and session $s' \neq s$ are not correlated. In particular, the assumptions on the errors are

$$E\left[\epsilon_{rgs}|x_{rgs}\right] = 0 \tag{A.2}$$

$$Var\left(\epsilon_{rgs}\right) = \sigma_{rr,gs} \tag{A.3}$$

$$Cov\left(\epsilon_{rgs}, \epsilon_{r'gs}\right) = \sigma_{rr',gs}$$
 if $r \neq r'$ (A.4)

$$Cov\left(\epsilon_{rgs}, \epsilon_{rg's}\right) = 0$$
 if $g \neq g'$ (A.5)

$$Cov\left(\epsilon_{rgs}, \epsilon_{rgs'}\right) = 0$$
 if $s \neq s'$ (A.6)

The corresponding formula for the robust covariance matrix (Cameron et al., 2011) in the linear regression model is given by

$$\widehat{\operatorname{Var}}\left(\widehat{\beta}\right) = \frac{N-1}{N-k} \left(\frac{G}{G-1}\right) (X'X)^{-1} \left(\sum_{s=1}^{S} \sum_{g=1}^{G_s} u'_{gs} u_{gs}\right) (X'X)^{-1}, \quad (A.7)$$

where $N = \sum_{s=1}^{S} \sum_{g=1}^{G_s} R_{gs}$ is the total number of observations, k = 2the number of regressors, $G = \sum_{s=1}^{S} G_s$ the total number of groups, Xthe $(N \times 2)$ -matrix of regressors, $u_{gs} = \sum_{r=1}^{R_{gs}} \epsilon_{rgs} x_{rgs}$, $\epsilon_{rgs} = y_{rgs} - x'_{rgs}\hat{\beta}$, $x_{rgs} = [1, \text{treatment}_{rgs}]$ the (2×1) -vector of independent variables for the observation in round r of group g in session s, and $\hat{\beta} = [\hat{\beta}_0, \hat{\beta}_1]$ the (2×1) vector of coefficient estimates.

We estimate (A.1) using logit regression when looking at the incidence of supra-competitive prices and the incidence of price coordination. We use linear regression when looking at market prices. All statistical tests reported in Section 5 are for the relevant no-treatment effect versus the two-sided alternative using z-tests with logit regression or t-tests with linear regression.

Since there are within matches some negative time trends in the data, we also estimate the following regression model with clustered standard errors at the group level to test for differences in the variable y between treatments a and b:

$$y_{rgs} = \gamma_0 + \gamma_1 \text{treatment}_{gs} + \gamma_2 \text{round}_{gs} + \epsilon_{rgs}, \qquad (A.8)$$

where γ_0 , γ_1 and γ_2 are coefficients to be estimated, ϵ_{rgs} are normally distributed errors, treatment_{gs} is a dummy that equals 1 if group g in session s participated in treatment a and 0 otherwise, and round_{gs} indicates in which round group g in session s participates. The significance level of γ_1 is relevant whether there is a difference between treatments. As these results are qualitative the same to the results of (A.1) we do not report them in this paper.²⁸

A.2 Individual Level

We also report on a few results on the individual level. At the level of the follower we are interested in the percentage of followers which set a lower/higher price than the leader. At the level of the leader we are interested in how often the leader sets a certain price, and the average price set. Differences between treatments are tested in a pairwise fashion as in (A.1), but the errors are clustered at the individual level instead of the group level.

 $^{^{28}\}mathrm{Results}$ are available from the author upon request.

B Session Details

Table B.1 provides detailed information on each session. We performed a binomial goodness-of-fit test to test the hypothesis that the continuation probability was binomially distributed with a 70% probability of continuation. The hypothesis was not rejected (rejection probability of 30.8%).

Treatment					Ma	tch		Earnings				
-	1	2	3	4	5	6	7	8	9	total	average mi	n max
SimEqual	4	7	1	1	3	3	3	1	1	24	€8.65 €8.0	0 €9.40
SimEqual	3	2	7	1	2	10	2	2	1	30	€10.44 €9.1	5 €13.10
SimUnequal	1	4	4	3	4	3	1	6	3	29	€9.47 €7.4	5 €11.30
SimUnequal	1	5	9	1	2	1	2	7	2	30	€9.47 €7.6	5 €11.40
Follower30	5	6	4	1	1	1	1	2	1	22	€9.24 €7.5	5 €11.65
Follower30	1	7	4	4	3	7	5	9	1	41	€12.64 €10.1	0 €18.10
Follower50	16	2	3	15	1	3	5	2	-	47	€17.73 €11.7	0 €24.60
Follower50	1	9	2	8	1	3	4	3	4	35	€13.88 €10.1	5 €16.95
Follower70	6	5	4	3	1	2	1	1	1	24	€10.68 €7.3	0 €14.95
Follower70	4	6	15	4	5	3	1	3	-	41	€13.58 €9.0	0 €20.65

Table B.1: Number of rounds played during each match, and average, minimum and maximum earnings, for all sessions.

There were 18 participants in each session. A dash '-' in match 9 indicates that the corresponding session had only 16 participants.

C Distribution of Prices per Treatment

Tables C.1 up to C.5 provide information per treatment on the distribution of prices chosen by every type of subject across all rounds and active groups.

Subject 1	Subject 2													
	3	4	5	6	7	8	9	10	11	12				
3	16.1	8.8	6.0	2.5	3.5	1.4	0.8	0.4	0.4	0.2	29.0			
4		3.1	6.0	3.7	1.0	1.4	0.6	0.2	0.0	1.2	11.9			
5			4.3	7.4	4.1	1.2	0.0	0.0	0.2	2.1	19.1			
6				4.3	3.9	0.8	0.8	0.8	0.2	1.4	14.8			
7					1.0	0.8	1.4	0.6	0.0	1.9	10.5			
8						0.2	0.4	0.2	0.0	1.2	5.1			
9							0.0	0.0	0.0	0.6	1.4			
10								0.0	0.0	0.0	1.2			
11									0.2	0.0	0.6			
12										2.3	6.2			
Total	16.1	11.9	16.3	17.9	13.6	6.0	4.1	2.3	1.0	10.9	100.0			

Table C.1: Distribution (%) of prices chosen in SIMEQUAL across all rounds and active groups (N=486).

Subject 30%				Sul	bject	70%					Total
	3	4	5	6	7	8	9	10	11	12	
3	24.3	5.5	2.1	1.7	0.6	0.2	0.0	0.0	0.2	0.9	35.4
4	5.3	5.5	4.1	2.3	1.5	0.2	0.4	0.2	0.4	0.4	20.2
5	1.7	1.7	3.8	3.0	2.3	0.0	0.4	0.4	0.2	0.4	13.8
6	1.1	2.1	3.2	2.8	1.9	0.8	0.2	0.2	0.2	0.4	12.8
7	1.5	1.1	1.3	1.1	0.6	0.6	0.6	0.6	0.0	0.2	7.5
8	0.8	0.4	0.6	0.9	0.6	0.4	0.0	0.0	0.0	0.2	3.8
9	0.4	0.0	0.4	0.2	0.2	0.0	0.0	0.0	0.0	0.2	1.3
10	0.2	0.2	0.6	0.6	0.2	0.0	0.0	0.0	0.0	0.2	1.9
11	0.2	0.2	0.4	0.4	0.0	0.0	0.0	0.0	0.0	0.0	1.1
12	1.1	0.2	0.4	0.0	0.0	0.2	0.0	0.2	0.0	0.2	2.3
Total	36.5	16.8	16.8	13.0	7.7	2.3	1.5	1.5	0.9	3.0	100.0

Table C.2: Distribution (%) of prices chosen in SIMUNEQUAL across all rounds and active groups (N=531).

Table C.3: Distribution (%) of prices chosen in FOLLOWER30 across all rounds and active groups (N=567).

Leader					Foll	ower					Total
	3	4	5	6	7	8	9	10	11	12	
3	39.0	3.5	0.4	0.2	0.7	0.9	0.5	0.7	0.2	7.2	53.3
4	4.1	2.3	0.2	0.0	0.0	0.0	0.0	0.0	0.0	0.5	7.1
5	0.0	4.4	1.9	0.2	0.2	0.0	0.0	0.0	0.0	0.4	7.1
6	0.0	0.0	3.4	0.7	0.0	0.0	0.0	0.0	0.0	0.0	4.1
7	0.0	0.0	0.0	2.7	2.3	0.0	0.0	0.0	0.0	0.2	5.1
8	0.0	0.0	0.0	0.0	0.9	0.7	0.2	0.0	0.0	0.0	1.8
9	0.0	0.0	0.0	0.0	0.0	0.9	0.5	0.0	0.0	0.0	1.4
10	0.0	0.0	0.0	0.0	0.0	0.0	1.6	0.4	0.0	0.0	1.9
11	0.0	0.0	0.0	0.0	0.0	0.0	0.0	3.2	1.1	0.4	4.6
12	0.0	0.0	0.0	0.0	0.0	0.2	0.0	0.2	9.0	4.4	13.8
Total	43.0	10.2	5.8	3.7	4.1	2.7	2.8	4.4	10.2	13.1	100.0

Leader					Folle	ower					Total
	3	4	5	6	7	8	9	10	11	12	-
3	23.2	1.5	0.3	0.0	0.0	0.0	0.1	0.4	0.0	3.9	29.4
4	1.7	3.6	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.3	5.6
5	0.3	1.3	3.9	0.1	0.1	0.1	0.0	0.1	0.0	0.3	6.4
6	0.0	0.0	0.9	4.1	0.0	0.0	0.0	0.0	0.0	0.6	5.5
7	0.0	0.0	0.0	0.9	3.2	0.0	0.0	0.0	0.0	0.1	4.2
8	0.0	0.0	0.0	0.0	0.4	3.2	0.0	0.0	0.0	0.0	3.6
9	0.0	0.0	0.0	0.0	0.0	0.9	3.6	0.1	0.0	0.0	4.6
10	0.0	0.0	0.0	0.0	0.0	0.0	0.7	4.9	0.1	0.0	5.8
11	0.0	0.0	0.0	0.0	0.0	0.0	0.0	1.7	2.5	0.0	4.2
12	0.1	0.0	0.0	0.0	0.0	0.0	0.0	0.1	4.9	25.5	30.7
Total	25.3	6.4	5.1	5.1	3.8	4.2	4.5	7.5	7.5	30.7	100.0

Table C.4: Distribution (%) of prices chosen in FOLLOWER50 across all rounds and active groups (N=691).

Table C.5: Distribution (%) of prices chosen in FOLLOWER70 across all rounds and active groups (N=544).

Leader					Foll	ower					Total
	3	4	5	6	7	8	9	10	11	12	
3	19.9	2.2	0.6	0.2	0.4	0.6	0.4	0.7	1.3	3.7	29.8
4	5.7	2.2	1.1	0.0	0.0	0.0	0.0	0.4	0.4	0.9	10.7
5	0.0	4.0	4.4	0.4	0.2	0.0	0.0	0.0	0.0	0.7	9.7
6	0.0	0.0	4.6	4.0	0.7	0.2	0.0	0.0	0.0	1.1	10.7
7	0.0	0.0	0.2	2.4	1.8	0.4	0.0	0.0	0.4	0.2	5.3
8	0.2	0.0	0.0	0.2	2.4	2.6	0.4	0.0	0.4	0.4	6.4
9	0.0	0.0	0.0	0.0	0.0	0.9	1.1	0.0	0.0	0.2	2.2
10	0.0	0.0	0.0	0.0	0.0	0.0	2.0	2.4	1.1	0.0	5.5
11	0.0	0.0	0.0	0.0	0.0	0.0	0.2	3.7	2.6	1.3	7.7
12	0.0	0.0	0.0	0.0	0.0	0.0	0.0	0.2	4.2	7.5	12.0
Total	25.7	8.5	10.9	7.2	5.5	4.6	4.0	7.4	10.3	16.0	100.0

D Instructions Follower70

Market decision making

You are going to participate in an experiment on market decision making.

We will first read the instructions aloud. Then you will have time to read them on your own. The instructions are identical for all participants. After reading, there is the possibility to ask questions individually. Please refrain from talking during the entire experiment.

The experiment consists of separate games. Each game has the same structure. You will play each game with a different person. You play at most once with the same person during the entire experiment. During one game you will play with the same player. Together, you and that other person form a group. You will never learn who the other player is.

Before the experiment starts, we randomly determine whether you are an A-player or a B-player. During the entire experiment you will keep this role. An A-player will always play with a B-player, and vice versa.

In this experiment you can earn points. The number of points you earn depends on the decisions made by you and those made by the other player in your group. At the beginning of the experiment, you receive 75 points in your account. At the end of each game, the points that you earned in that game will be added to your account. At the end of the experiment the number of points in your account will be converted to euros, at a rate of $\notin 0.07$ per point.

The experiment is expected to last for approximately 75 minutes.

Rules of a Game

During a game you play with the same person. A game consists of several rounds. The number of rounds is random. After every round a number from 1, 2, 3, up to and including 100 is drawn by a computer. If the number is smaller than or equal to 70, a new round starts. If, however, it is higher than 70, the game ends. Thus, there is a probability of 70% that a new round of a game will be played and a probability of 30% that the game ends. If a

game ends a new game starts. A new game will be played with a different person. Hence, in each game you meet a new person.

Rules in a Round

Each round of a game consists of four steps. These steps are the same every round.

Step 1: pricing decision A-player

The A-player chooses one of the following prices:

Step 2: pricing decision B-player

After the A-player has chosen his price, the B-player learns the price chosen by the A-player in step 1. Next, the B-player chooses one of the following prices:

You and the other player receive the following number of points:

- If your price is lower than the price chosen by the other player, you receive a number of points equal to your price. The other player receives 0 points.
- If your price is the same as the price chosen by the other player, the A-player receives a number of points equal to 30% of his price. The B-player receives a number of points equal to 70% of his price.
- If your price is higher than the price chosen by the other player, you receive 0 points. The other player receives a number of points equal to his price.

The number of points you receive can be found in Table 1. This table is added to the instructions. Table 1 reads as follows. The possible prices of the A-player are indicated next to the rows. The possible prices of the Bplayer are indicated above the columns. In the cell at which row and column intersect, the number of points the A-player receives is up to the left and the number of points the B-player receives is down to the right.

Examples

- Suppose that the A-player chooses a price of 6, and the B-player chooses a price of 8. In Table 1 you move down until you reach the row which has 6 on the left of it. Then, you move to the column with 8 above it. You see that the A-player receives 6 points, while the B-player receives 0 points.
- Suppose that the A-player chooses a price of 9, and the B-player chooses a price of 9. In Table 1 you move down until you reach the row which has 9 on the left of it. Then, you move to the column with 9 above it. You see that the A-player receives 2.7 points, while the B-player receives 6.3 points.
- Suppose that the A-player chooses a price of 11, and the B-player chooses a price of 7. In Table 1 you move down until you reach the row which has 11 on the left of it. Then, you move to the column with 7 above it. You see that the A-player receives 0 points, while the B-player receives 7 points.

Please make sure you understand Table 1 and also make sure that it is in line with the instructions above.

Step 3: summary of round

After the B-player has chosen his price, the A-player learns the price chosen by the B-player in step 2. The number of points you have received will also be displayed. Throughout the experiment, there will also be a box on your screen where you can observe the prices chosen by you and the other player in previous rounds during a game.

Step 4: continuation outcome

The drawn number is displayed. Remember that if the number is smaller than or equal to 70, a new round of a game starts. If, however, it is higher than 70, the game ends.

End of experiment

After the last game has been played, the experiment ends. You receive a message on your screen that no further game will take place. At the end of the experiment, the total number of points in your account will be converted

at a rate of $\in 0.07$ per point. Before being paid in private, you have to hand in the instructions.

After the experiment, please do not discuss the content of the experiment with anyone, including people who did not participate.

Please refrain from talking throughout the experiment.

THANK YOU VERY MUCH FOR PARTICIPATING AND GOOD LUCK!

Table 1

									Pri	ice cł	iosen	by I	3-pla	yer							
			3	4	4	į	5	(6	1	7	٤	3		9		10	1	1	1	2
	3	0.9		3		3		3		3		3		3		3		3		3	
	5		2.1		0		0		0		0		0		0		0		0		0
	4	0		1.2		4		4		4		4		4		4		4		4	
	-1		3		2.8		0		0		0		0		0		0		0		0
	Б	0		0		1.5		5		5		5		5		5		5		5	
ye.	0		3		4		3.5		0		0		0		0		0		0		0
ola	6	0		0		0		1.8		6		6		6		6		6		6	
-	0		3		4		5		4.2		0		0		0		0		0		0
Y -	7	0		0		0		0		2.1		7		7		7		7		7	
ē.	'		3		4		5		6		4.9		0		0		0		0		0
sen	8	0		0		0		0		0		2.4		8		8		8		8	
100	0		3		4		5		6		7		5.6		0		0		0		0
G	9	0		0		0		0		0		0		2.7		9		9		9	
ić	0		3		4		5		6		7		8		6.3		0		0		0
Pr	10	0		0		0		0		0		0		0		3		10		10	
	10		3		4		5		6		7		8		9		7		0		0
	11	0		0		0		0		0		0		0		0		3.3		11	
	11		3		4		5		6		7		8		9		10		7.7		0
	12	0		0		0		0		0		0		0		0		0		3.6	
	14		3		4		5		6		7		8		9		10		11		8.4

References

- ALBUQUERQUE, ANA (2009): "Peer Firms in Relative Performance Evaluation," Journal of Accounting and Economics, 48(1), 69–89.
- ARGENTON, CÉDRIC, AND WIELAND MÜLLER (2012): "Collusion in Experimental Bertrand Duopolies with Convex Costs: The Role of Cost Asymmetry," International Journal of Industrial Organization, 30(6), 508–517.
- BIGONI, MARIA, SVEN-OLOF FRIDOLFSSON, CHLOE LE COQ, AND GIAN-CARLO SPAGNOLO (2012): "Fines, Leniency and Rewards in Antitrust," *RAND Journal of Economics*, 43(2), 368–390.
- BLONSKI, MATTHIAS, PETER OCKENFELS, AND GIANCARLO SPAGNOLO (2011): "Equilibrium Selection in the Repeated Prisoner's Dilemma: Axiomatic Approach and Experimental Evidence," *American Economic Journal: Microeconomics*, 3(3), 164–192.
- CAMERER, COLIN F. (2003): Behavioral Game Theory: Experiments on Strategic Interaction. Princeton University Press, Princeton, New Jersey.
- CAMERON, A. COLIN, JONAH B. GELBACH, AND DOUGLAS L. MILLER (2011): "Robust Inference with Multiway Clustering," Journal of Business & Economic Statistics, 29(2), 238–249.
- CASON, TIMOTHY N., SAU-HIM PAUL LAU, AND VAI-LAM MUI (2013): "Learning, Teaching, and Turn Taking in the Repeated Assignment Game," *Economic Theory*, 54(2), 335–357.
- DAL BÓ, PEDRO (2005): "Cooperation under the Shadow of the Future: Experimental Evidence from Infinitely Repeated Games," American Economic Review, 95(5), 1591–1604.
- DAL BÓ, PEDRO, AND GUILLAUME R. FRÉCHETTE (2011): "The Evolution of Cooperation in Infinitely Repeated Games: Experimental Evidence," *American Economic Review*, 101(1), 411–429.
- DATTA MAGO, SHAKUN, AND EMMANUEL DECHENAUX (2009): "Price Leadership and Firm Size Asymmetry: An Experimental Analysis," *Experimental Economics*, 12(3), 289–317.

- DAVIS, DOUGLAS D., AND BART J. WILSON (2002): "Collusion in Procurement Auctions: An Experimental Examination," *Economic Inquiry*, 40(2), 213–230.
- DG COMPETITION (2006): "Case COMP/F-1/38.121 Fittings," http://ec.europa.eu/competition/antitrust/cases/dec_docs/ 38121/38121_1007_1.pdf, Accessed: April 26, 2013.
- (2007): "Case COMP/38.432 Professional Videotape," http://ec.europa.eu/competition/antitrust/cases/dec_docs/ 38432/38432_526_5.pdf, Accessed: April 26, 2013.
- (2008): "Case COMP/39181 Candle Waxes," http: //ec.europa.eu/competition/antitrust/cases/dec_docs/39181/ 39181_1908_8.pdf, Accessed: April 26, 2013.
- DUFWENBERG, MARTIN, AND URI GNEEZY (2000): "Price Competition and Market Concentration: An Experimental Study," *International Journal* of Industrial Organization, 18(1), 7–22.
- DUGAR, SUBHASIS, AND ARNAB MITRA (2009): "The Size of the Cost Asymmetry and Bertrand Competition: Experimental Evidence," Unpublished manuscript.

(2013): "Bertrand Competition with Asymmetric Marginal Costs," Unpublished manuscript.

- FEHR, ERNST, AND KLAUS M. SCHMIDT (1999): "A Theory of Fairness, Competition, and Cooperation," *Quarterly Journal of Economics*, 114(3), 817–868.
- FISCHBACHER, URS (2007): "z-Tree: Zurich Toolbox for Ready-Made Economic Experiments," *Experimental Economics*, 10(21), 171–178.
- FONSECA, MIGUEL A., STEFFEN HUCK, AND HANS-THEO NORMANN (2005): "Playing Cournot although They shouldn't: Endogenous Timing in Experimental Duopolies with Asymmetric Cost," *Economic Theory*, 25(3), 669–677.
- FONSECA, MIGUEL A., WIELAND MÜLLER, AND HANS-THEO NORMANN (2006): "Endogenous Timing in Duopoly: Experimental Evidence," International Journal of Game Theory, 34(3), 443–456.

- FONSECA, MIGUEL A., AND HANS-THEO NORMANN (2008): "Mergers, Asymmetries and Collusion: Experimental Evidence," *The Economic Journal*, 118(527), 387–400.
- (2012): "Explicit vs. Tacit Collusion The Impact of Communication in Oligopoly Experiments," *European Economic Review*, 56(8), 1759–1772.
- FRIEDMAN, JAMES W. (1971): "A Non-Cooperative Equilibrium for Supergames," *Review of Economic Studies*, 38(1), 1–12.
- GIBBONS, ROBERT, AND KEVIN J. MURPHY (1990): "Relative Performance Evaluation for Chief Executive Officers," *Industrial and Labor Relations Review*, 43(3), 30S–51S.
- HILDENBRAND, ANDREAS (2010): "Cournot or Stackelberg Competition? A Survey of Experimental Research," MPRA Paper No. 26845, Munich Personal RePEc Archive.
- HUCK, STEFFEN, WIELAND MÜLLER, AND HANS-THEO NORMANN (2001): "Stackelberg beats Cournot: On Collusion and Efficiency in Experimental Markets," *The Economic Journal*, 111(474), 749–765.
- (2002): "To Commit or Not to Commit: Endogenous Timing in Experimental Duopoly Markets," *Games and Economic Behavior*, 38(2), 240–264.
- IVALDI, MARC, BRUNO JULLIEN, PATRICK REY, PAUL SEABRIGHT, AND JEAN TIROLE (2003): "The Economics of Tacit Collusion," Report for DG Competition, European Commission.
- KÜBLER, DOROTHEA, AND WIELAND MÜLLER (2002): "Simultaneous and Sequential Price Competition in Heterogeneous Duopoly Markets: Experimental Evidence," *International Journal of Industrial Organization*, 20(10), 1437–1460.
- MASON, CHARLES F., OWEN R. PHILLIPS, AND CLIFFORD NOWELL (1992):
 "Duopoly Behavior in Asymmetric Markets: An Experimental Evaluation," *The Review of Economics and Statistics*, 74(4), 662–670.
- MOTTA, MASSIMO (2004): Competition Policy Theory and Practice. Cambridge University Press, New York.

- MOURAVIEV, IGOR, AND PATRICK REY (2011): "Collusion and Leadership," International Journal of Industrial Organization, 29(6), 705–717.
- OOSTERBEEK, HESSEL, RANDOLPH SLOOF, AND GIJS VAN DE KUILEN (2004): "Cultural Differences in Ultimatum Game Experiments: Evidence from a Meta-Analysis," *Experimental Economics*, 7(2), 171–188.
- PHILLIPS, OWEN R., DALE J. MENKHAUS, AND JOHN N. THUROW (2011):
 "The Small Firm in a Quantity Choosing Game: Some Experimental Evidence," *Review of Industrial Organization*, 38(2), 191–207.
- PUZZELLO, DANIELA (2008): "Tie-Breaking Rules and Divisibility in Experimental Duopoly Markets," Journal of Economic Behavior & Organization, 67(1), 164–179.
- ROTH, ALVIN E. (1995): "Bargaining Experiments," in *The Handbook of Experimental Economics*, ed. by J. H. Kagel, and A. E. Roth. Princeton University Press.
- ROTH, ALVIN E., AND J. KEITH MURNIGHAN (1978): "Equilibrium Behaviour and Repeated Play of the Prisoner's Dilemma," *Journal of Mathematical Psychology*, 17(2), 189–198.
- SCHERER, FREDERIC M., AND DAVID ROSS (1990): Industrial Market Structure and Economic Performance. Houghton Mifflin, Boston, 3rd edn.
- WANG, ZHONGMIN (2009): "(Mixed) Strategy in Oligopoly Pricing: Evidence from Gasoline Price Cycles before and under a Timing Regulation," *Journal of Political Economy*, 117(6), 987–1030.

List of research reports

groningen

12001-HRM&OB: Veltrop, D.B., C.L.M. Hermes, T.J.B.M. Postma and J. de Haan, A Tale of Two Factions: Exploring the Relationship between Factional Faultlines and Conflict Management in Pension Fund Boards

12002-EEF: Angelini, V. and J.O. Mierau, Social and Economic Aspects of Childhood Health: Evidence from Western-Europe

12003-Other: Valkenhoef, G.H.M. van, T. Tervonen, E.O. de Brock and H. Hillege, Clinical trials information in drug development and regulation: existing systems and standards

12004-EEF: Toolsema, L.A. and M.A. Allers, Welfare financing: Grant allocation and efficiency

12005-EEF: Boonman, T.M., J.P.A.M. Jacobs and G.H. Kuper, The Global Financial Crisis and currency crises in Latin America

12006-EEF: Kuper, G.H. and E. Sterken, Participation and Performance at the London 2012 Olympics

12007-Other: Zhao, J., G.H.M. van Valkenhoef, E.O. de Brock and H. Hillege, ADDIS: an automated way to do network meta-analysis

12008-GEM: Hoorn, A.A.J. van, Individualism and the cultural roots of management practices

12009-EEF: Dungey, M., J.P.A.M. Jacobs, J. Tian and S. van Norden, On trend-cycle decomposition and data revision

12010-EEF: Jong-A-Pin, R., J-E. Sturm and J. de Haan, Using real-time data to test for political budget cycles

12011-EEF: Samarina, A., Monetary targeting and financial system characteristics: An empirical analysis

12012-EEF: Alessie, R., V. Angelini and P. van Santen, Pension wealth and household savings in Europe: Evidence from SHARELIFE

13001-EEF: Kuper, G.H. and M. Mulder, Cross-border infrastructure constraints, regulatory measures and economic integration of the Dutch - German gas market

13002-EEF: Klein Goldewijk, G.M. and J.P.A.M. Jacobs, The relation between stature and long bone length in the Roman Empire

13003-EEF: Mulder, M. and L. Schoonbeek, Decomposing changes in competition in the Dutch electricity market through the Residual Supply Index

13004-EEF: Kuper, G.H. and M. Mulder, Cross-border constraints, institutional changes and integration of the Dutch – German gas market

13005-EEF: Wiese, R., Do political or economic factors drive healthcare financing privatisations? Empirical evidence from OECD countries

13006-EEF: Elhorst, J.P., P. Heijnen, A. Samarina and J.P.A.M. Jacobs, State transfers at different moments in time: A spatial probit approach

13007-EEF: Mierau, J.O., The activity and lethality of militant groups: Ideology, capacity, and environment

13008-EEF: Dijkstra, P.T., M.A. Haan and M. Mulder, The effect of industry structure and yardstick design on strategic behavior with yardstick competition: an experimental study

13009-GEM: Hoorn, A.A.J. van, Values of financial services professionals and the global financial crisis as a crisis of ethics

13010-EEF: Boonman, T.M., Sovereign defaults, business cycles and economic growth in Latin America, 1870-2012

13011-EEF: He, X., J.P.A.M Jacobs, G.H. Kuper and J.E. Ligthart, On the impact of the global financial crisis on the euro area

13012-GEM: Hoorn, A.A.J. van, Generational shifts in managerial values and the coming of a global business culture

13013-EEF: Samarina, A. and J.E. Sturm, Factors leading to inflation targeting – The impact of adoption

13014-EEF: Allers, M.A. and E. Merkus, Soft budget constraint but no moral hazard? The Dutch local government bailout puzzle

13015-GEM: Hoorn, A.A.J. van, Trust and management: Explaining cross-national differences in work autonomy

13016-EEF: Boonman, T.M., J.P.A.M. Jacobs and G.H. Kuper, Sovereign debt crises in Latin America: A market pressure approach

13017-GEM: Oosterhaven, J., M.C. Bouwmeester and M. Nozaki, The impact of production and infrastructure shocks: A non-linear input-output programming approach, tested on an hypothetical economy

13018-EEF: Cavapozzi, D., W. Han and R. Miniaci, Alternative weighting structures for multidimensional poverty assessment

14001-OPERA: Germs, R. and N.D. van Foreest, Optimal control of production-inventory systems with constant and compound poisson demand

14002-EEF: Bao, T. and J. Duffy, Adaptive vs. eductive learning: Theory and evidence

14003-OPERA: Syntetos, A.A. and R.H. Teunter, On the calculation of safety stocks

14004-EEF: Bouwmeester, M.C., J. Oosterhaven and J.M. Rueda-Cantuche, Measuring the EU value added embodied in EU foreign exports by consolidating 27 national supply and use tables for 2000-2007



university of groningen

14006-EEF: Reijnders, L.S.M., The college gender gap reversal: Insights from a life-cycle perspective

14007-EEF: Reijnders, L.S.M., Child care subsidies with endogenous education and fertility

14008-EEF: Otter, P.W., J.P.A.M. Jacobs and A.H.J. den Reijer, A criterion for the number of factors in a data-rich environment

14009-EEF: Mierau, J.O. and E. Suari Andreu, Fiscal rules and government size in the European Union

14010-EEF: Dijkstra, P.T., M.A. Haan and M. Mulder, Industry structure and collusion with uniform yardstick competition: theory and experiments

14011-EEF: Huizingh, E. and M. Mulder, Effectiveness of regulatory interventions on firm behavior: a randomized field experiment with e-commerce firms

14012-GEM: Bressand, A., Proving the old spell wrong: New African hydrocarbon producers and the 'resource curse'

14013-EEF: Dijkstra P.T., Price leadership and unequal market sharing: Collusion in experimental markets

14014-EEF: Angelini, V., M. Bertoni, and L. Corazzini, Unpacking the determinants of life satisfaction: A survey experiment

14015-EEF: Heijdra, B.J., J.O. Mierau, and T. Trimborn, Stimulating annuity markets

14016-GEM: Bezemer, D., M. Grydaki, and L. Zhang, Is financial development bad for growth?

14017-EEF: De Cao, E. and C. Lutz, Sensitive survey questions: measuring attitudes regarding female circumcision through a list experiment

14018-EEF: De Cao, E., The height production function from birth to maturity

14019-EEF: Allers, M.A. and J.B. Geertsema, The effects of local government amalgamation on public spending and service levels. Evidence from 15 years of municipal boundary reform

14020-EEF: Kuper, G.H. and J.H. Veurink, Central bank independence and political pressure in the Greenspan era

14021-GEM: Samarina, A. and D. Bezemer, Capital flows and financial intermediation: is EMU different?

14022-EEF: Soetevent, A.R. and L. Zhou, Loss Modification Incentives for Insurers Under ExpectedUtility and Loss Aversion



14023-EEF: Allers, M.A. and W. Vermeulen, Fiscal Equalization, Capitalization and the Flypaper Effect.

14024-GEM: Hoorn, A.A.J. van, Trust, Workplace Organization, and Comparative Economic Development.

14025-GEM: Bezemer, D., and L. Zhang, From Boom to Bust in de Credit Cycle: The Role of Mortgage Credit.

14026-GEM: Zhang, L., and D. Bezemer, How the Credit Cycle Affects Growth: The Role of Bank Balance Sheets.

14027-EEF: Bružikas, T., and A.R. Soetevent, Detailed Data and Changes in Market Structure: The Move to Unmanned Gasoline Service Stations.

14028-EEF: Bouwmeester, M.C., and B. Scholtens, Cross-border Spillovers from European Gas Infrastructure Investments.

14029-EEF: Lestano, and G.H. Kuper, Correlation Dynamics in East Asian Financial Markets.

14030-GEM: Bezemer, D.J., and M. Grydaki, Nonfinancial Sectors Debt and the U.S. Great Moderation.

14031-EEF: Hermes, N., and R. Lensink, Financial Liberalization and Capital Flight: Evidence from the African Continent.

14032-OPERA: Blok, C. de, A. Seepma, I. Roukema, D.P. van Donk, B. Keulen, and R. Otte, Digitalisering in Strafrechtketens: Ervaringen in Denemarken, Engeland, Oostenrijk en Estland vanuit een Supply Chain Perspectief.

14033-OPERA: Olde Keizer, M.C.A., and R.H. Teunter, Opportunistic condition-based maintenance and aperiodic inspections for a two-unit series system.

14034-EEF: Kuper, G.H., G. Sierksma, and F.C.R. Spieksma, Using Tennis Rankings to Predict Performance in Upcoming Tournaments

15001-EEF: Bao, T., X. Tian, X. Yu, Dictator Game with Indivisibility of Money

15002-GEM: Chen, Q., E. Dietzenbacher, and B. Los, The Effects of Ageing and Urbanization on China's Future Population and Labor Force

15003-EEF: Allers, M., B. van Ommeren, and B. Geertsema, Does intermunicipal cooperation create inefficiency? A comparison of interest rates paid by intermunicipal organizations, amalgamated municipalities and not recently amalgamated municipalities

15004-EEF: Dijkstra, P.T., M.A. Haan, and M. Mulder, Design of Yardstick Competition and Consumer Prices: Experimental Evidence

15005-EEF: Dijkstra, P.T., Price Leadership and Unequal Market Sharing: Collusion in Experimental Markets

www.rug.nl/feb